

Master's Programme in Mathematics and Operations Research

Modeling the Role of Battery Storage in the Nordic Energy System

A 10-Year Perspective on Renewable Integration and Capacity Expansion

Rohit Kumar

© 2024

This work is licensed under a Creative Commons "Attribution-NonCommercial-ShareAlike 4.0 International" license.





Author Rohit Kumar

Title Modeling the Role of Battery Storage in the Nordic Energy System — A 10-Year Perspective on Renewable Integration and Capacity Expansion

Degree programme Mathematics and Operations Research

Major Applied Mathematics

Supervisor Prof. Fabricio Oliviera

Advisor M.Sc. (Tech.) Helmi Hankimaa

Date 11 November 2024	Number of pages 49	Language English
-----------------------	--------------------	------------------

Abstract

The transition to a sustainable energy system is a pressing global challenge, and battery energy storage system (BESS) are emerging as a promising solution to improve renewable energy integration and grid stability. This thesis examines the integration of BESS into the Nordic energy system between 2023 and 2033, focusing on their role in improving renewable energy adoption and grid performance. The study aims to evaluate the potential of BESS investments in addressing variability in renewable generation and optimizing system costs. Utilizing the EMPIRE model, a cost-based energy system optimization tool, the study evaluates investments in BESS and renewable capacities to explore their impact on overall system costs. The analysis reveals a highly uneven adoption of BESS throughout the region, with Sweden leading because of its diverse mix of renewable energy. In contrast, other countries such as Norway, Finland, and Denmark rely on alternative flexible solutions such as hydro storage, gas technologies, and existing interconnections. The results highlight the complex interplay between storage, renewable resources, and existing infrastructure, emphasizing that the path to a sustainable energy system is shaped by unique national circumstances. Although BESS is shown to be effective in mitigating renewable curtailment and supporting grid reliability, its limited adoption in certain countries suggests that broader strategies are equally crucial. The study acknowledges limitations, such as static assumptions for renewable generation and demand, which provide opportunities for further research to refine energy models and explore complementary flexibility options. While the work primarily provides a case study of BESS integration in the Nordic region, it offers a starting point for assessing storage technologies within diverse energy contexts.

Keywords Battery Energy Storage System, Renewable Energy Integration, Nordic Energy System, Capacity Expansion Models, Battery Market Development, EMPIRE Model

Contents

At	ostrac	rt (3
Co	onten	ts	4
Sy	mbol	s and Abbreviations	5
1	Intr	oduction	6
2	Lite	rature review	8
	2.1	Capacity Expansion Models	8
	2.2	Long-Term Bottom-Up Models	9
		2.2.1 TIMES	9
		2.2.2 MESSAGE	10
		2.2.3 E2M2	11
		2.2.4 DIMENSION	13
3	EM	PIRE Model	15
	3.1	Scenario Structure	16
	3.2	Optimization Model	17
	3.3	Modeled Technologies	17
	3.4	Model Inputs	18
	3.5	Model Outputs	19
	3.6	Comparison with other Models	20
4	Nor	dic Adaptation of EMPIRE	21
	4.1	Node Structure	21
	4.2	Time Horizon	22
	4.3	Input Data for the Nordic Energy System	23
	4.4	Offshore Wind Farm Clusters	26
	4.5	Scenario Data	28
5	Rest	ılts	30
	5.1	BESS Deployment	30
		5.1.1 BESS Capacity in 2033	30
		5.1.2 Limited BESS Investments in Other Nordic Countries	30
	5.2	Renewable Energy Investments across the Nordics	32
	5.3	Energy Mixes across the Nordics	34
	5.4	Curtailment and Flexibility Needs	36
	5.5	Transmission Analysis Across the Nordic Region	38
	5.6	Summary	41
6	Con	clusion	42

Symbols and Abbreviations

Symbols

CO_2	carbon dioxide
tCO_2	tonne of CO_2
MtCO ₂ eq	megatonnes of CO ₂ -equivalent
kg CO ₂ /GJ	kilograms of CO ₂ per gigajoule
€/tCO ₂	euros per tonne of CO_2
MW	megawatts
MWh	megawatt-hours
GWh	gigawatt-hours
MW/MWh	power-to-energy ratio
€/kW	euros per kilowatt
€/MW	euros per megawatt
€/MWh	euros per megawatt-hour
€/GJ	euros per gigajoule
€/MW-km	euros per megawatt-kilometer
km	kilometers

Abbreviations

BESS	battery energy storage system		
CCGT	combined cycle gas turbine		
CCS	carbon capture and storage		
CEMs	capacity expansion models		
DIMENSION	Dispatch and Investment Model for European Electricity Markets		
DIME	Dispatch and Investment Model for Electricity Markets in Europe		
DSM	Demand Side Management		
E2M2	European Electricity Market Model		
EMPIRE	European Model for Power system Investment with Renewable Energy		
HVAC	high-voltage alternating current		
HVDC	high-voltage direct current		
IEA	International Energy Agency		
MESSAGE	Model for Energy Supply Strategy Alternatives and their General		
	Environmental Impact		
NEA	Nuclear Energy Agency		
NREAP	National Renewable Energy Action Plan		
OCGT	open cycle gas turbine		
TIMES	The Integrated MARKAL-EFOM System		
VRE	variable renewable energy		
WCSS	Within-Cluster Sum of Squares		

1 Introduction

Due to the concerning and growing consequences of climate change, there has been a worldwide shift towards the usage of renewable energy sources. Recent years have been the warmest on record, with global average temperatures rising 1.1°C above pre-industrial levels by 2019 [1]. The consequences, consisting of extreme weather events, underscore the urgent need for sustained action to safeguard global well-being [1].

The European Union (EU) has established itself as a leader in response to this struggle. The European Green Deal aims to achieve climate neutrality for the EU by 2050 [1]. The interim objective is to reduce greenhouse gas emissions by 55% by 2030 in comparison to 1990 levels [1]. In accordance with this ambitious goal, the Nordic countries have established their own rigorous climate objectives. Finland targets climate neutrality by 2035, Denmark targets a 70% reduction in emissions by 2030, Norway aims for a 55% reduction by 2030, and Sweden plans a 63% reduction by 2030, with climate neutrality by 2045 [2].

Battery energy storage system (BESS) has become essential for incorporating renewable energy into power systems, aiding in grid stabilization by holding energy during peak output and releasing it when demand surpasses supply [3]. Recent improvements in lithium-based batteries, recognized for their high energy density and extended longevity, are crucial for large-scale applications such as electric vehicles and renewable energy storage [3].

However, the growth of the BESS market in the Nordic region has been uneven. Sweden and Finland have led in deploying grid-scale battery systems, with Sweden expecting over 400 MW for operation in 2024 and Finland anticipating over 300 MW of grid-scale batteries in the next two years [4]. Meanwhile, Norway, despite its ambitions to lead the battery storage market, has focused on hydro resources for long-duration storage, falling behind in BESS deployments [4]. In Sweden, where the demand for flexibility services may soon outpace the pace of battery installations, this rapid development has raised concerns about potential market saturation [4].

Given the rapid expansion of BESS and the potential risks of overcapacity, capacity expansion models are essential for optimizing the planning of generation, storage, and transmission infrastructure. These models aim to minimize total system costs over long-term horizons, ensuring that investments in technologies such as BESS align with renewable energy targets while considering system constraints [5].

In this context, a variety of models have been developed to assist in long-term energy system planning. For example, TIMES [6] and MESSAGE [7] focus on minimizing long-term costs but face challenges in representing short-term constraints. Different models, such as DIMENSION and E2M2, are better at short-term dispatch and market interactions, but they are not as good at cross-sectoral integration or stochastic analysis [8]. As renewable energy and storage technologies become more prevalent, these models need to evolve to balance long-term planning with the uncertainties of short-term decisions.

Despite these developments, these models frequently abstract away complexities because they are simplified depictions of the real world. This could lead to different results depending on the modeling method used, especially in systems that are saturated with variable renewable energy (VRE) [5]. This abstraction may result in deficiencies when modeling specific technologies such as BESS, which necessitate a thorough examination of their operational dynamics and market relations.

The EMPIRE model (European Model for Power System Investment with Renewable Energy) [9] helps address some of the limitations seen in other capacity expansion models by combining long-term investment planning with short-term operational strategies, particularly under high renewable energy penetration scenarios. While it does not fully resolve all challenges, EMPIRE provides a more detailed framework for modeling the interactions between renewable energy variability and energy storage technologies. The aim of this thesis is to adapt the EMPIRE model to estimate the development of the battery market in the Nordics between 2023 and 2033, with five investment periods occurring every two years. This time frame was selected because rapid growth is expected within the battery market, with possible saturation within the next decade [4]. By adopting the EMPIRE model, this research aims to provide insights into the role of battery storage in the Nordic energy landscape, focusing on how investment and operational decisions may evolve.

This thesis is structured as follows: Chapter 2 reviews existing capacity expansion models and their purpose, Chapter 3 details the EMPIRE model, Chapter 4 outlines the methodology for adapting the model for the Nordic region, Chapter 5 discusses the results from the analysis of the Nordic dataset, and Chapter 6 summarises the findings.

2 Literature review

This literature review will explore modeling approaches used to evaluate long-term energy system investments, with a specific focus on the integration of BESS. To understand how BESS can support the transition to renewable energy in the Nordic region, the review will first introduce capacity expansion models and explain their importance in long-term strategic planning. The discussion will then differentiate between top-down and bottom-up models, emphasizing the relevance of long-term bottom-up models for this study. Key models in this sector will be reviewed to highlight their strengths and limitations. This review will identify gaps in these existing models, demonstrating the need to adapt the EMPIRE model to better capture the role of BESS in the Nordic energy system.

2.1 Capacity Expansion Models

Capacity expansion models, or CEMs, are essential tools used to plan and optimize energy systems over long periods. These models assist decision-makers in determining the most cost-effective approach to constructing and expanding energy infrastructure, encompassing generation, storage, and transmission systems [10]. CEMs are crucial for understanding how current decisions will impact future energy systems, often projecting decades ahead [10].

Policymakers and energy planners rely on CEMs to simulate different scenarios and strategies, taking into account factors such as fluctuating fuel prices, changes in technology, and evolving energy policies [10]. Through these simulations, CEMs help identify the optimal mix of energy resources and offer insights into how power systems can evolve over time [11].

The classification of CEMs has evolved over time. [12] laid the groundwork by differentiating models based on factors such as top-down versus bottom-up approaches, as well as their focus on either short- or long-term planning horizons [13]. This framework was enhanced by adding more details about the model's purpose, its mathematical structure, its focus on geography, and its sectoral coverage [13, 14]. Later, the conversation shifted toward hybrid models, blending both bottom-up and top-down approaches [13, 15]. Further refinements were made in 2013, introducing a more balanced classification system that gave equal weight to all the key characteristics identified earlier [13, 16]. [13, 17] present a thorough review of 13 tools that focused on community-scale energy systems. The review looked at data input types, the supply technologies, and the demand-side management and storage methods [13, 17]. These ongoing efforts to refine CEM classifications underscore the increasing complexity of energy systems and the need for models that can address both technological innovations and policy challenges.

CEMs can be broadly classified into two types based on their analytical approach: top-down and bottom-up models [13, 18]. Top-down models analyze the relationship between the energy sector and the broader economy. This allows for the assessment of the socio-economic impacts of energy and climate policies, including effects on employment, public welfare, and economic growth [13]. These models provide a

macroeconomic perspective, focusing on how changes in the energy system influence the overall economy rather than specific technological components [13]. However, this often limits their ability to assess sector-specific or technology-specific policies, making them less effective for analyzing energy technology choices [18].

In contrast, bottom-up models offer a more detailed and technology-oriented analysis of energy systems. These models focus on individual technologies and how they interact across various energy sectors, making them suitable for evaluating the integration of renewable energy sources and energy storage systems [13, 18]. While bottom-up models are valuable at providing insights into the technical and economic performance of energy technologies, they may not fully capture broader economic impacts, such as employment or gross domestic product changes, which are more typically addressed by top-down models [18].

Within bottom-up models, a further distinction can be made between short-term and long-term models [13]. Short-term bottom-up models prioritize operational aspects, such as real-time electricity dispatch and balancing supply and demand on an hourly or daily basis [13]. This helps in addressing immediate challenges such as grid stability and flexibility [13]. In contrast, long-term bottom-up models are better suited for strategic planning and evaluating decisions on infrastructure investments such as generation, storage, and transmission over extended time horizons [13]. Given the focus of this thesis on integrating BESS into the Nordic energy system, a long-term bottom-up model is a better choice for understanding how BESS investments will aid renewable energy integration and impact the system over time.

2.2 Long-Term Bottom-Up Models

Long-term bottom-up models are used to project the evolution of energy systems over extended periods. These models focus on capacity expansion planning, particularly through infrastructure investments, which is important for meeting future energy demands. By identifying optimal investment strategies, these models ensure long-term reliability and sustainability. Their emphasis on infrastructure development provides key insights for transitioning to low-carbon energy sources.

In contrast to short-term models that emphasize market dynamics or immediate technology adoption, long-term bottom-up models offer an enhanced perspective on strategic planning, particularly in evaluating the infrastructure required to integrate VRE sources. Their strength lies in addressing large-scale challenges, making them well-suited to analyze the Nordic energy system's transformation.

The following sections will review prominent long-term bottom-up models, examining their structure and potential for analyzing renewable energy integration.

2.2.1 TIMES

The TIMES (The Integrated MARKAL-EFOM System) [6] model is a flexible, bottomup, optimization-based model designed to represent energy system dynamics over a multi-period time horizon. The MARKAL (MARKet ALlocation) and EFOM (Energy Flow Optimization Model) models served as inspiration for TIMES, which simulates the ideal configuration of generation, storage, reserve capacity, and transmission infrastructure at the lowest total system cost. Its scope encompasses decisions around equipment investment, operation, energy trade, and fuel sourcing while meeting criteria such as emissions limits or policy goals. The model includes mechanisms to manage reserve capacity. One such mechanism is the peaking reserve constraint, which ensures sufficient production capacity to meet peak demand with a user-defined margin. This accounts for uncertainties such as equipment failures or demand surges. Technologies are assigned specific coefficients to determine their contribution to peak capacity. Reliable power plants can contribute fully, while variable technologies may have reduced contributions. This approach ensures that the model can balance short-term supply and demand while maintaining system reliability. [6]

The model also models the interactions between energy producers and consumers, seeking to establish supply-demand equilibrium. Market imperfections such as taxes, subsidies, and constraints can be incorporated to simulate policy interventions (such as emissions regulations), which makes the model highly adaptable to different regulatory contexts. [6]

The model operates on a flexible time structure, divided into user-defined periods spanning several years, allowing it to simulate long-term developments in the energy system. It separates data inputs from the time horizon, enabling users to adjust time frames without extensive recalibration, thus simplifying the assessment of long-term trends and scenarios. One of TIMES' strengths lies in its use of time slices, where each year is subdivided into smaller units, such as seasons, weekdays, and portions of the day. This aspect of the model is valuable for handling technologies sensitive to temporal variations, such as wind turbines or electricity storage. By modeling sub-annual time slices, the model can balance supply and demand in short-term intervals and ensure that peak loads are met. Technologies such as pumped storage or night storage devices are optimized to shift energy across time slices, which ensures that reserve capacity is available during peak demand periods. [6]

TIMES has been applied extensively in various regional and national contexts. The TIMES-Norway model was used to investigate hydrogen's role in decarbonizing an energy system transitioning from petroleum [19]. This study explored different levels of hydrogen production and consumption, assessing their impacts on system costs, emissions, and interactions with other energy sources [19]. In another application, VEDA-TIMES was used to model Indonesia's path to a 100% renewable energy system by 2050 [20]. This variant utilized an hourly resolution to analyze the role of energy storage in managing the variability of renewable sources such as wind and solar [20]. Furthermore, the MIRET-EU variant of the TIMES model was employed to assess hydrogen's contribution to achieving net-zero emissions in Europe by 2050 [21]. This study generated projections of hydrogen production, consumption, and imports across various sectors, examining policy scenarios and technology options [21].

2.2.2 MESSAGE

The Model for Energy Supply Strategy Alternatives and their General Environmental Impact (MESSAGE) [7], developed by the International Institute for Applied Systems Analysis (IIASA), is a dynamic linear programming model that minimizes total energy supply costs over a specific time horizon. It finds the best balance between secondary energy needs (such as heating and electricity) and primary energy supplies (such as fossil fuels and renewables), taking into account things such as the availability of resources, the effects on the environment, and the rate at which new technologies are adopted [7].

MESSAGE uses a cost-minimization objective function to guide energy infrastructure investment decisions. It optimizes the addition of generation capacity under various resource and technological constraints, ensuring that investment strategies meet energy demand at the lowest possible cost [7]. These investment decisions are distributed across a dynamic time horizon, enabling long-term energy planning [7].

In addition to its cost-effectiveness, MESSAGE is flexible and adaptable, making it suitable for various regional contexts and global scenarios. The model incorporates load regions and distinguishes between domestic and imported resources, allowing it to optimize energy systems at national and international scales [7]. It also accounts for environmental constraints by simulating emissions of greenhouse gases and pollutants such as sulfur dioxide (SO₂) and nitrogen oxides (NO_x), making it a valuable tool for evaluating environmental policies alongside economic objectives [8].

MESSAGE also accounts for resource availability and depletion over time, influencing the shift in investment toward alternative technologies as costs rise with scarcity [7]. It considers the life cycle of infrastructure and imposes build-up constraints, preventing rapid, unrealistic technology deployment and reflecting the real-world challenges of scaling energy infrastructure [7].

The model's time horizon, divided into periods typically spanning 5 to 10 years, allows for long-term simulations of up to 120 years. This design allows MESSAGE to simulate gradual technological advancements, changes in resource availability, and environmental impacts over time. As a result, it is highly effective for strategic energy planning, including generation, storage, and transportation systems [7].

MESSAGE has been extensively used to support the development of global energy transition pathways such as those for the World Energy Council [22]. It has also played a key role in generating greenhouse gas emission scenarios for the Intergovernmental Panel on Climate Change [23]. Applications of MESSAGE include a wide range of studies, such as scenario assessments aimed at climate stabilization [24, 25], evaluations of innovation programs in the Iranian electricity sector [26], and the assessment of policy options for enhancing renewable energy adoption [27]. Additionally, the model has been used to analyze energy supply strategies in the Baltic states and to develop sustainable energy plans for Cuba [28, 29].

2.2.3 E2M2

The European Electricity Market Model (E2M2) [30], developed by the Institute of Energy Economics and Rational Energy Use (IER) at the University of Stuttgart, is a tool for optimizing and simulating electricity markets, with a particular focus on Europe. It operates under the assumption of a perfectly competitive market, where firms are price takers, and all participants have full access to information [31].

Under these conditions, E2M2 simultaneously optimizes long-term investment in new generation capacity and short-term unit commitment (scheduling and dispatching power generation to meet demand at minimum system cost) decisions [30]. The model also incorporates representations of thermal power plants and renewable energy sources. It includes flexibility solutions such as demand side management (DSM), energy storage systems, and power-to-heat technologies [30].

The primary objective of E2M2 is to minimize the system's total cost while meeting electricity demand. These costs include investments in new power plants, operational and maintenance costs, fuel costs, and carbon dioxide (CO_2) certificate costs [30]. Fixed costs of existing capacities are treated as sunk. However, the model captures the operating costs of both new and existing plants, including start-up costs that depend on the unit's minimum load during start-up [30]. This cost structure ensures the model accurately reflects the real operational conditions of power plants.

E2M2 uses important factors such as demand for electricity and heat, generation capacities, techno-economic parameters, and regulatory factors such as fuel prices and CO_2 emission limits as inputs [30]. It calculates marginal electricity prices using the dual variable of the power balance equation [30]. This provides insights into day-ahead market prices. It also estimates CO_2 certificate prices when emissions limits are imposed [32].

Key restrictions, such as system adequacy, govern E2M2. It ensures that electricity and heat demand are met while maintaining sufficient reserve capacity. The reserve requirement accounts for potential plant outages and the variability of renewable generation [32]. Technical constraints on thermal plants, such as ramping rates and minimum operating times, are incorporated using a mixed-integer linear programming approach. This allows for more precise unit commitment decisions [32]. The model also accounts for district heating systems, particularly combined heat and power (CHP) plants, which provide both electricity and heat [32].

E2M2 handles uncertainties in renewable energy generation using a stochastic framework. For example, it models wind power variability with a scenario recombining tree that simulates different wind conditions over time [32]. This approach allows for more resilient planning compared to deterministic models that assume perfect foresight.

The E2M2 model has been effectively applied to analyze and optimize electricity markets in various contexts. One notable application is its use in modeling the German electricity market [32]. In this context, the model simulates unit commitment and investment decisions while considering factors specific to Germany, such as district heating systems and net electricity exchange with neighboring countries [32]. Additionally, the model has been employed in a coupled modeling framework to identify and quantify the "efficiency gap" in electricity systems [30]. The "efficiency gap" here refers to the difference between the actual performance of electricity systems and their theoretically optimal performance under ideal conditions. Researchers achieved a more realistic picture of how well the system works and the associated costs by combining E2M2 with the agent-based market model for the investigation of renewable and integrated energy systems (AMIRIS) [33]. This model simulates the decisions that individual market actors make. This approach was chosen to account

for deviations from perfect competition and market distortions [30].

2.2.4 DIMENSION

The Institute of Energy Economics at the University of Cologne (EWI) created the DIMENSION [34] model, a linear optimization tool meant to simulate the future development of European electricity markets, particularly under increasing integration of renewable energy sources. It aims to provide decision-makers with insights into the complex dynamics of power generation and system expansion in a market that must adapt to both conventional and renewable technologies. [34]

The model's optimization approach minimizes the total costs of the electricity system. These costs include variable costs, investment costs, maintenance expenses, and costs related to ramping power plants. It achieves this cost minimization while balancing electricity demand and supply within the system and simulating capacity constraints of both generation and transmission infrastructure. The model allows for investments in new generation capacity and evaluates storage technologies that can mitigate the variability of renewable energy sources. Additionally, it incorporates CHP plants by reflecting their technical features, such as their power-to-heat ratio, and simulates their role in the future energy mix. [34]

The model represents the electricity system using a graph structure where nodes represent different system components such as power plants, demand regions, and storage facilities. The connections between these nodes simulate the flow of electricity, enabling DIMENSION to evaluate how constraints such as transmission line capacities affect overall system performance. This structure is beneficial in providing a proper structure for simulating the operation of the electricity system under various constraints and future scenarios. [34]

A distinct feature of the model is its simulation of DSM and the role of electric vehicles (EVs) as virtual power storage, where EVs are modeled as flexible resources that can be charged when electricity demand is low and discharged when demand is high. By simulating these flexibility options, DIMENSION provides valuable insights into how future electricity systems could integrate new technologies while managing the increasing share of renewable energy. [34]

A limitation of the model is its exogenous treatment of renewable energy, requiring users to manually input the share and mix of renewable sources rather than allowing the model to optimize these values. The developers recognize that enabling the model to make these decisions would result in more accurate and insightful simulations. [34]

The DIME [35] model, a precursor to DIMENSION, has been applied in various research and consulting projects. DIME was used in energy policy analysis for the German government to evaluate scenarios for achieving climate targets [36, 37]. It also informed studies on integrating renewable energy for the German Energy Agency [38, 39] and assessed the impact of renewable deployment on conventional power markets [40]. Although these applications refer to DIME, they remain relevant for understanding the broader capabilities that DIMENSION inherits.

Despite their diverse applications, these models share inherent limitations in addressing the complexities of modern energy systems. TIMES and MESSAGE are

powerful for long-term energy planning, but they only allow study of broad strategic goals and lack precise operational insights needed to comprehend short-term variations and technology-specific dynamics. DIMENSION and E2M2 both focus on long-term power system investment decisions and lack short-term operational flexibility, especially in high-renewable scenarios. Both models struggle to reflect the changing interactions between renewable energy sources and storage systems across temporal scales. These limitations place restrictions on modern energy systems that involve extensive modeling of renewable integration and operational flexibility.

To address these limitations, we have selected EMPIRE, which enables both longterm strategic planning and the short-term operational flexibility needed to analyze interactions between renewables and storage systems across temporal scales.

3 EMPIRE Model

The EMPIRE model [41] is a linear optimization model developed to optimize both investments and operations in power systems over a long planning horizon. Implemented as an open-source package, it includes a scenario generation procedure that allows users to explore various energy futures and analyze investment strategies under uncertain conditions. A core feature of EMPIRE is its formulation as a multihorizon stochastic program, which handles uncertainty in VRE sources [41]. By generating different scenarios for renewable generation and electricity demand, the model allows users to evaluate a range of investment strategies [41]. This enables decision-makers to account for short-term operational needs alongside long-term strategic goals in an energy system shaped by fluctuating renewable supply and demand.

Figure 1 provides an overview of the EMPIRE model structure. It shows the model's key components: inputs, the multi-horizon stochastic optimization model, and outputs. The maps below the model structure illustrate possible transmission expansions and generation mixes across different regions, reflecting spatial optimization outcomes.



Figure 1: Overview of the EMPIRE model[9].

The subsequent sections will explore the components of the EMPIRE model in detail. Section 3.1 describes how the model handles uncertainty through its multi-horizon stochastic scenario structure. Section 3.2 follows by detailing how the optimization model minimizes both investment and operational costs while accounting for system constraints and uncertainties. Afterward, Section 3.4 provides an overview of the data required to simulate the system, covering the economic, technical, and stochastic inputs. These inputs feed into the model's decision-making process, leading to Section 3.5, which outlines the key results generated. Finally, Section 3.6 concludes with a comparison of EMPIRE with the other reviewed models.

3.1 Scenario Structure

The EMPIRE model is formulated as a multi-horizon stochastic program to handle uncertainty in the energy system. This approach integrates investment decisions with operational decisions across various operational scenarios, allowing the model to manage the interactions between strategic planning and the variability of renewable energy sources.

In EMPIRE, strategic decisions are made under the assumption of perfect foresight, meaning that the model assumes full knowledge of future variables such as fuel prices, technology costs, and demand growth when making long-term investments. Additionally, it is assumed that operational decisions within each period do not affect future strategic or operational decisions.[41]

Figure 2 illustrates the structure of a multi-horizon stochastic program that integrates both strategic and operational uncertainties. Each decision node represents a branching point where long-term strategic choices are made, which influence subsequent operational decisions. As one advances through various stages, uncertainties arise, leading to multiple potential future scenarios. This framework illustrates the complexity of decision-making across various horizons, where earlier strategic choices can limit or enhance future operational actions.

EMPIRE operates across two types of scenarios: investment scenarios and operational scenarios. Investment scenarios focus on long-term planning. Uncertainties such as technological advancements, regulatory changes, or market conditions could influence future strategies. Operational scenarios focus on short-term variations in load profiles and generation from wind, solar, and seasonal hydroelectric sources. Each investment period is evaluated against a set of operational scenarios, ensuring long-term strategies account for potential short-term conditions. [41]

To reduce computational complexity, EMPIRE employs temporal aggregation. Investment decisions are grouped into broader time blocks, minimizing the number of decision points required for long-term planning. For operational scenarios, a subset of representative hours is used to model system operations. They focus on regular and extreme load seasons to capture critical periods without needing to simulate every hour of the year. [41]



Figure 2: Multi-horizon stochastic program with strategic and operational uncertainty in EMPIRE [41].

3.2 Optimization Model

The optimization model in EMPIRE aims to minimize total system costs over a long planning horizon, which forms the objective function. This includes both investment and operational costs, where the decision variables focus on expanding generation, storage, transmission infrastructure, and operating energy resources [41].

On the operational time scale, the decision variables govern the system's hourly functions, such as dispatching generation units, operating storage systems, and coordinating regional electricity transfers. The model incorporates various factors, such as transmission and storage efficiency losses and load shedding, to realistically capture energy flows and maintain the balance between supply and demand. [41]

The model operates under several key constraints that guide its decision-making. Investment constraints limit how much generation capacity and transmission infrastructure can be added each year, ensuring that expansions occur gradually over time. Power balance constraints ensure that electricity supply always meets demand at every node on an hourly basis, maintaining system stability. Generation constraints regulate how much energy each unit can produce, with additional restrictions such as ramp-up limits for thermal generators, which control how quickly their output can change. Transmission constraints cap the electricity that can flow between regions based on the capacity of transmission lines and account for transmission losses. Reservoir water availability and seasonal energy constraints both limit hydroelectric generation, ensuring effective year-round use of water resources. Lastly, storage constraints manage the operation of storage systems by controlling their energy levels and accounting for efficiency losses during charging and discharging. [41]

3.3 Modeled Technologies

The model incorporates a variety of technologies that contribute to system operations and investment dynamics. These technologies can be grouped into conventional generation, renewable generation, nuclear, hydro, bio, waste-to-energy, storage, and transmission.

Conventional generation technologies include coal, lignite, oil, and gas, which provide dispatchable power to support baseload and peak demands. Among these, open cycle gas turbine (OCGT) is primarily used for peaking, while combined cycle gas turbine (CCGT) is favored for its efficiency in sustained generation. Additionally, carbon capture and storage (CCS) is applied to certain conventional fuels, capturing CO_2 emissions to reduce their environmental impact.

Renewable generation technologies, including wind (both onshore and offshore), solar, and geothermal, play a significant role in decarbonization. Offshore wind, in particular, offers high-capacity renewable energy, though subject to variability, while solar complements other sources by providing power during daylight hours. Geothermal energy, by contrast, is a stable source unaffected by weather, enhancing system reliability.

Nuclear power contributes low-carbon electricity, supporting long-term sustainability targets and providing a reliable power supply to the grid. Meanwhile, hydroelectric power includes both reservoir-based and run-of-the-river systems. Reservoir-based hydro provides flexibility and storage potential, while run-of-the-river hydro contributes steady renewable generation without storage capacity.

Bioenergy and waste-to-energy technologies add further flexibility to the system. As dispatchable renewable options, they not only provide power but also address waste management, allowing renewable generation that can be adjusted to meet demand.

Energy storage technologies in the model include hydro pumped storage and lithium-ion BESS. Hydro pump storage offers rapid dispatch capabilities, essential for balancing peak demands, while lithium-ion BESS manages variability in renewable generation by storing excess power and releasing it during high-demand periods.

Lastly, the transmission infrastructure, which consists of high voltage alternating current (HVAC) overhead lines and high voltage direct current (HVDC) cables, is crucial for moving electricity across regions. HVDC cables are particularly suited for long-distance, high-efficiency transmission, facilitating the delivery of renewable energy where it is most needed.

3.4 Model Inputs

The model relies on a range of inputs to capture the energy system's economic, technical, and stochastic dynamics. These inputs are organized into three categories, each valuable in enabling the model to simulate the financial and operational complexities of power systems.

Economic parameters are essential for evaluating the financial feasibility of different system configurations. These inputs include discount rates, generation and transmission investment costs, operational expenses, and load-shedding costs [41]. The model also imposes limits on maximum investments in capacity and transmission infrastructure [41]. By accounting for both upfront capital expenditures and ongoing operational costs, EMPIRE enables cost-benefit analyses of various energy system investments [41].

Technology features represent the technical characteristics of the energy system's components. These include aggregated loads and generation capacities for each country, along with seasonal capacity factors for different generation technologies [41]. Generator-specific attributes, such as efficiency and heat rates, are also incorporated [41]. Additionally, transmission losses between regions and pump efficiencies for energy storage systems are considered [41]. These inputs aid in modeling energy flows and operational constraints, enabling EMPIRE to optimize system performance under realistic technical conditions.

Lastly, stochastic scenarios serve as inputs that simulate the uncertain nature of renewable generation and demand. The model generates hourly data series for parameters such as electricity demand, wind power (onshore and offshore), solar, and hydroelectric production [41]. This random scenario generation keeps the statistical properties of the original data, which means that correlations between variables such as load and renewable generation are kept [41].

The scenario generation process begins by selecting a random year from the available dataset. For each season, the model selects a starting hour and uses consecutive hours to populate the seasonal data, preserving temporal relationships within the data. Extreme load seasons, characterized by peak demand or low renewable output, are also modeled to assess system performance during stress. These scenarios ensure that both system-wide and regional events are represented accurately. Once generated, the scenarios are validated to ensure alignment with the statistical properties of the original data. [41]

3.5 Model Outputs

The EMPIRE model generates various outputs that provide both strategic and operational insights into the energy system. These outputs help identify optimal investment paths for the system and how such a system operates under different scenarios.

The primary outputs are the investment decisions regarding generation capacity and transmission line expansions [41]. These decisions are made for each investment period, which spans multiple years. For each period, the model indicates how much new capacity should be installed and where upgrades to the transmission network are required to ensure that future demand can be met [41].

On the operational level, the model provides hourly system operation outputs [41]. These include the dispatch schedules for generation units, which show how much electricity each unit should produce during each hour in each operating scenario to meet demand [41]. Additionally, the model tracks power flows between nodes, showing how electricity is transmitted between different regions [41]. This is particularly important for understanding how the grid manages renewable energy variability and demand changes across different areas.

The model also outputs information on energy used for pumping, which refers to the amount of electricity used to operate pumped storage facilities. These facilities store energy by pumping water to higher elevations during periods of low demand and then releasing it to generate electricity during peak demand periods. By tracking this, the model helps assess the role of storage in counteracting the variability in the production of VRE resources. [41]

Another key operational output is load shedding, which occurs when the system cannot meet demand and customers experience power outages [41]. This output reflects how well the system maintains reliability under different scenarios, and minimizing load shedding is a critical objective for ensuring a stable energy supply.

Lastly, the model produces the total system costs, which include both the costs of investments in new infrastructure and the operational costs incurred during each period [41]. These costs are essential for evaluating the economic feasibility of different energy system setups and strategies.

3.6 Comparison with other Models

The EMPIRE model has numerous advantages over other frequently used CEMs. While DIMENSION incorporates short-term dispatch decisions, it is a deterministic model that disregards short-term uncertainty. This makes it less appropriate for systems that exhibit significant variability in renewable energy sources.

Another model that incorporates operational decisions is E2M2. However, it optimizes investments in single steps, which restricts its ability to completely evaluate the long-term effects of those decisions across multiple periods. It also lacks the mechanisms necessary to manage the variability of renewable energy and the balancing needs required to maintain system stability, due to its emphasis on electricity market simulations.

The TIMES framework incorporates a two-stage stochastic program that enables strategic decision-making under uncertainty, thereby addressing both short- and longterm dynamics. Nevertheless, its two-stage structure does not permit modelling multi-period investment decisions. Thus, it cannot be used for analyses that require ongoing modifications to long-term strategies as uncertainties evolve.

Despite its value in the optimization of long-term system costs, MESSAGE does not provide a complete representation of short-term operational constraints. Its focus is primarily on strategic investments, which limits its ability to fully integrate the operational dynamics needed for systems with significant renewable energy penetration and energy storage requirements.

In contrast, EMPIRE is the optimal choice for the analysis of renewable energy systems with significant storage and flexibility requirements. The reasoning behind this choice is its ability to seamlessly integrate investment decisions with granular modeling of operational decisions. EMPIRE offers a superior framework for assessing both short-term and long-term strategies in the presence of uncertainty by incorporating the operational uncertainties associated with renewable energy generation and the technical characteristics of various technologies.

The subsequent section provides an in-depth explanation of the specific methods and data that were employed to apply the EMPIRE model to the Nordic region, following the detailed discussion of the model's structure. The section outlines the methodology in which key parameters were defined and how they affect the model's results by concentrating on the Nordic dataset.

4 Nordic Adaptation of EMPIRE

The input data for this study is composed of several files and scenario information, which guide the overall functioning of the EMPIRE model. This data includes country-specific parameters, technology characteristics, transmission links, and seasonal scaling for renewable energy resources. Figure 3 provides a visual representation of the overall data flow, showing how raw user input and scenario-generated data are converted into optimization-ready model inputs. In the following subsections, the data used to model the Nordic energy system will be explained in detail.



Figure 3: Input data structure and flow in the EMPIRE model [9].

4.1 Node Structure

The model forms a total of 33 nodes to represent different regions, countries, and offshore wind across Europe, as shown in Table 1. The focus is on the Nordic region, which include Finland, Sweden, Norway, and Denmark. These nodes represent key electricity markets and allow for a thorough analysis of energy flows, generation, and storage within the Nordic energy system. Each country, except Finland, is divided into multiple price areas (SE1-SE4 for Sweden, NO1-NO5 for Norway, and DK1-DK2 for Denmark), reflecting regional differences in electricity prices.

To capture the interactions and dependencies of the Nordic energy system, the model also incorporates nodes for other European countries that are closely linked through transmission lines. Countries such as Germany, Great Britain, Belgium, Poland, the Netherlands, Estonia, Latvia, and Lithuania are included due to their transmission connections to the Nordic grid, which allow for energy exchanges. High-voltage direct current interconnectors link these regions to the Nordic countries, allowing the model to capture cross-border energy flows and accurately represent the interactions between interconnected European energy systems.

Furthermore, specific offshore wind farm clusters have been defined to capture the growing role of offshore wind energy, particularly in the North Sea. These clusters are linked to the transmission systems of the nodes and provide renewable energy resources to the interconnected grid. The offshore wind farm clusters are discussed in further detail in Section 4.4.

Country Nodes	Offshore Wind Farms Nodes	
Belgium	Cluster1	
DK1	Cluster2	
DK2	Cluster3	
Estonia	Cluster4	
Finland	Cluster5	
Germany	Cluster6	
Great Britain	Cluster7	
Latvia	Cluster8	
Lithuania	Cluster9	
Netherlands	Cluster10	
NO1	Cluster11	
NO2	Cluster12	
NO3	Cluster13	
NO4		
NO5		
Poland		
SE1		
SE2		
SE3		
SE4		

Table 1: Nodes part of the Nordic dataset.

4.2 Time Horizon

The model evaluates the system's operation over a 10-year horizon, spanning from 2023 to 2033, with operational impacts extending through 2035 to capture the full effect of each investment. Divided into five 2-year investment periods, each period provides an assessment of how strategic investments modify the system to adapt to future demand. Investments are assumed to become immediately available, with no delay or construction time required, allowing resources to be operational as soon as each investment period begins.

4.3 Input Data for the Nordic Energy System

This study utilizes a structured set of input data specific to the Nordic energy system, supporting the EMPIRE model's simulation of infrastructure and performance.

A structured summary of each input parameter's function in the model is given in Table 2. It includes crucial details such as cost metrics, capacity constraints, and operational characteristics for transmission, storage, and generation that are necessary for modeling and assessing the Nordic energy system. All data used in the Nordic EMPIRE model is sourced from publicly available datasets, reports, and literature, ensuring transparency and replicability.

Data Category	Description	Source
CO ₂ Price	CO ₂ prices for each period, expressed in euros per tonne of CO ₂ $(€/tCO_2)$.	[42]
Capital Costs	Capital costs for generator technolo- gies , expressed in euros per kilowatt (€/kW)	[43][44]
Fixed Operation and Mainte- nance Costs	Annual operation and maintenance costs for generator technologies, given in euros per kilowatt (€/kW).	[43][44]
Variable Operation and Main- tenance Costs	Costs per unit of electricity pro- duction for generator technologies, specified in euros per megawatt-hour (€/MWh).	[43]
Efficiency	Percentage of fuel converted to elec- tricity for generator technologies, ex- pressed as a percentage (%).	[43]
Operational Lifespan	Expected operational lifespan for generator technologies, measured in years.	[43]
Fuel Costs	Fuel Costs, specified in euros per gigajoule (€/GJ).	[42][45]
CCS Costs	Cost of capturing and storing CO_2 per tonne, given in euros per tonne of CO_2 (C/tCO_2).	[46]
Initial Generation Capacity	Initial generation capacity of tech- nologies for each node, measured in megawatts (MW).	[47][48][49]
Retirement Scale Factor	Percentages of initial capacity that will retire in each investment period for each generation technology.	Linear ex- trapolation

Table 2: Inputs used in the EMPIRE model and data sources of the Nordic dataset.

Continuation of Table 2			
Data Category	Description	Source	
Resource Limits on Installed Capacity	Resource limits for maximum in- stalled capacity of generators, speci- fied in megawatts (MW).	[50][51][52]	
Ramp Rate	Rate of output change per hour for thermal generators, given as a per- centage of installed capacity per hour (%).	IEA,NEA	
Generator Availability	Percentage of installed capacity available at any given hour.	IEA,NEA	
CO ₂ Emissions Content	CO_2 emissions per unit of fuel con- sumed, specified in kilograms of CO_2 per gigajoule (kg CO_2/GJ).	[53]	
Annual Electricity Demand	Annual electricity demand for each node, given in megawatt-hours (MWh).	[54][55]	
Cost of Lost Load	Economic cost associated with un- met electricity demand, specified in euros per megawatt-hour (€/MWh).	[56]	
Maximum Hydro Production	Maximum expected annual produc- tion from hydro generators, for reservoir-based and run-of-the-river systems, specified in megawatt- hours (MWh).	NREAP	
Initial Power Capacity	Initial charging/discharging capac- ity for each node, divided by hydro pump storage or lithium-ion BESS, measured in megawatts (MW).	[47] [57]	
Power Capital Cost	Cost required to install storage capacity (€/kW).	[58] [59]	
Power Fixed Operation and Maintenance Cost	Fixed operation and maintenance cost per year for maintaining storage capacity (€/kW).	[59]	
Power Maximum Built Capac- ity	Maximum additional storage capac- ity that can be built per period (MW).	Assumption	
Energy Capital Cost	Cost required to add energy storage capacity (€/kWh).	[58] [59]	
Energy Fixed Operation and Maintenance Cost	Fixed operation and maintenance cost per year for energy storage (\mathbb{C}/kWh) .	Default: 0	

Continuation of Table 2			
Data Category	Description	Source	
Energy Initial Capacity	Initial energy storage capacity for each node, divided by hydro pump storage or lithium-ion BESS, given in megawatt-hours (MWh).	[47] [57]	
Energy Maximum Built Capacity	Maximum additional energy storage capacity that can be added per period (MWh).	Assumption	
Energy Maximum Installed Capacity	Maximum total energy storage ca- pacity allowed for each node and stor- age type, including a 10% increase over existing capacity (MWh).	Assumption	
Power Maximum Installed Capacity	Maximum allowable storage capac- ity for charging/discharging built per period,including a 50% increase over installed capacity, measured in megawatts (MW).	Assumption	
Storage Initial Energy Level	Initial energy level of storage systems as a percentage of their installed en- ergy capacity (%).	Assumption	
Storage Charging Efficiency	Efficiency of energy storage during the charging process, expressed as a percentage (%).	Assumption	
Storage Discharging Efficiency	Efficiency of energy release during the discharging process, expressed as a percentage (%).	Assumption	
Storage Power to Energy Ratio	Required ratio of installed power to energy storage capacity for depen- dent storage systems (MW/MWh).	Assumption	
Storage Self-Discharge Effi- ciency	Hourly percentage of energy loss (self-discharge) from storage systems (%).	Assumption	
Lifetime	Anticipated operational lifespan of storage systems (years).	[60]	
Transmission Line Efficiency	Percentage of electricity transmitted that reaches the destination node, expressed as a percentage (%).	Default: 0.97	

Continuation of Table 2		
Data Category	Description	Source
Maximum Transmission Ca- pacity Built	Maximum transmission capacity that can be constructed between nodes within a period, specified in megawatts (MW).	Assumption
Transmission Line Length	Physical length of the transmission line between nodes, measured in kilometers (km).	Distances defined by map
Transmission Capital Cost	Investment cost per MW-km for transmission lines, including HVAC overhead lines and HVDC cables, specified in euros per megawatt- kilometer (€/MW-km).	Default: 0
Transmission Operation and Maintenance Cost	Annual operation and maintenance cost for transmission lines, specified in euros per megawatt (€/MW).	Default: 0
Initial Transmission Capacity	Initial transmission capacity between nodes within a period, measured in megawatts (MW).	[61]
Transmission Line Lifespan	Expected operational lifespan of transmission lines between nodes, specified in years.	Default: 40 years

4.4 Offshore Wind Farm Clusters

The offshore wind farm data used in this study was obtained from the Global Wind Power Tracker [48]. This dataset provides information about wind farms around the world. Figure 4 depicts the 122 operational offshore wind farms considered for this analysis, located in the Nordic region and surrounding areas.

Given the large number of wind farms, it was necessary to group them into clusters for the sake of computational tractability. Modeling each wind farm as a distinct node would have significantly increased computational complexity, making the model potentially unmanageable or preventing it from solving. To overcome this, the k-means clustering algorithm was applied to group the wind farms based on their geographical proximity. This approach allowed for the inclusion of all wind farms in the model while maintaining a manageable number of nodes (see Figure 5).



Figure 4: Operational offshore wind farms considered in the study.

K-means clustering is an unsupervised machine learning algorithm used to partition a dataset into a predefined number of clusters, k. The algorithm works iteratively to assign data points to clusters based on their distance from the cluster centroids. The process begins by randomly selecting k data points as initial centroids. Each point is assigned to the cluster whose centroid is closest, calculated using Euclidean distance. The centroids are then recalculated based on the new cluster assignments, and this process repeats until the centroids stabilize. [62]

One challenge with k-means clustering is determining the optimal number of clusters, k. If too few clusters are chosen, the wind farms may be too geographically dispersed within a cluster, reducing model accuracy. Conversely, if too many clusters are used, the efficiency gained from clustering is reduced. The elbow method was used to identify a reasonable number of clusters, based on the Within-Cluster Sum of Squares (WCSS), which measures the variance within each cluster [62]. The goal is to minimize the WCSS while avoiding an excessive number of clusters [62].



Figure 5: Resulting clusters of offshore wind farms after applying k-means clustering.

The elbow method involves running the k-means algorithm for a range of k values and plotting the WCSS for each value of k [62]. As the number of clusters increases, the WCSS decreases, but the rate of decrease slows at a certain point, forming an "elbow" in the plot [62]. This point is taken to represent the optimal number of clusters, beyond which adding more clusters results in diminishing returns [62].

Figure 5 illustrates the 13 clusters identified in this study, determined using the elbow method. This number was chosen to provide clear geographical groupings while maintaining computational efficiency. This clustering approach allows the model to capture the distinct characteristics of different geographical areas without the complexity of representing each wind farm as an individual node.

It is important to note that the model does not generate new nodes when simulating future investments. This means that, while the model can expand capacity within the established clusters, it cannot create new offshore wind farms in previously undeveloped locations. Consequently, all new investments in offshore wind are constrained to the geographic areas where current wind farms are located, reflecting an assumption that future capacity expansions will occur within or near existing infrastructure.

4.5 Scenario Data

The scenario data used in this study originates from the Europe v51 dataset included in the EMPIRE code repository [63]. Table 3 summarizes the scenario data used in the model. This dataset provides hourly values for profiles such as electric load, hydro run-of-river, hydro seasonal, solar, wind onshore, and wind offshore for each applicable node, spanning from January 1, 2015, through December 31, 2019. To align the data with the study's timeline, values from 2015–2019 were reindexed to represent the years 2023–2027. Renewable energy profiles (wind, solar, and hydro), represented by capacity factors (the ratio of actual output to maximum potential output), were assumed to remain similar over time. This approach considers variations in renewable generation patterns from 2015–2019 to be representative of the period up to 2035.

For demand, the pattern of fluctuations (daily and seasonal variations) was assumed to remain unchanged, while the total demand level was adjusted to align with forecasts provided in the input data.

For countries with multiple price zones, including Sweden (SE1–SE4), Norway (NO1–NO5), and Denmark (DK1, DK2), the data was divided to represent each price zone accurately. The scaling factors used for profiles were assigned the same value across all price zones within a country, ensuring consistency. For numeric profiles, such as energy capacities or production figures, the values were simply divided among the zones according to the same ratios as the changes in hydropower production across those zones.

Name	Description
Electric Load Profile	Hourly electricity demand data for each node.
Wind Profile (Onshore)	Hourly wind generation data for onshore wind farms.
Wind Profile (Offshore)	Hourly wind generation data for offshore wind farms.
Solar Profile	Hourly Solar Generation Data.
Hydro Profile	Seasonal and hourly hydroelectric generation data.

Table 3: Overview of scenario data in the Nordic dataset.

5 Results

The results of this study provide insights into the projected development of BESS and renewable energy integration in the Nordic countries from 2023 to 2033. This section begins with an analysis of BESS capacity distribution across the Nordic region, which highlights the differences in storage adoption among the countries by 2033.

5.1 BESS Deployment

5.1.1 BESS Capacity in 2033

Figure 6 displays the projected BESS capacities in the Nordic countries for the operational period 2033-2035, with power capacities (MW) shown on the left and energy capacities (MWh) on the right. The model indicates that Sweden has the highest installed BESS capacities, both in terms of power and energy. In contrast, Norway, Finland, and Denmark show considerably lower levels of installed BESS capacity in both metrics.

The maps indicate that Sweden's installed energy capacity extends significantly beyond those of its neighboring countries, with a pronounced difference in scale.



Figure 6: Projected installed BESS capacities by 2033 in the Nordic region. The left panel shows installed power capacity (MW), while the right panel displays installed energy capacity (MWh).

5.1.2 Limited BESS Investments in Other Nordic Countries

Figure 7 illustrates the progression of BESS investments across the modeled periods. These plots highlight Sweden's dominance in BESS adoption, supported by significant capacity additions across all investment periods. In contrast, Norway exhibits negligible investments during the first four periods, similar to the patterns observed for Denmark and Finland throughout the entire modeled horizon. However, Norway displays a small but notable increase in both energy and power capacities in the final investment

period, marking its only measurable adoption of BESS. Denmark and Finland remain consistently below the threshold of 1 MW or 1 MWh, rendering their contributions insignificant for further analysis.



Figure 7: Total installed BESS capacity by investment period in each country. The top plot shows the progression of installed energy capacity (MWh), and the bottom plot shows the progression of installed power capacity (MW).

While Norway's BESS investments are minimal even with the inclusion of the the last operational period, it does align with its abundant hydro power storage capacity. Hydro power in Norway provides both energy storage and grid stability, dampening the need for large-scale battery investments and acting as a natural flexibility resource. Although temporal variations in Norway's hydro capacity are not visible in the plots, the capacities remain consistently high across all periods. Table 4 provides Norway's pumped hydro storage capacities for power and energy across the modeled periods.

Investment	Installed power	Installed energy
period	capacity (MW)	capacity (MWh)
2025-2027	1329.50	4908400.00
2027-2029	1329.50	4908400.00
2029-2031	1329.50	4908400.00
2031-2033	1329.50	4908400.00
2033-2035	1329.50	4908400.00

 Table 4: Norway's Pumped Hydro Storage Installed Capacities

5.2 Renewable Energy Investments across the Nordics

Figures 8, 9 and 10 illustrate the modeled progression of installed capacities for solar, wind, and run-of-river hydro across the Nordic countries. The plots provide a comparative view of how renewable energy sources evolve in each country.



Figure 8: Installed solar capacity by investment period in each country.



Figure 9: Installed wind (onshore and offshore) capacity by investment period in each country.





As seen in Figure 8, Sweden consistently leads the Nordic region in installed solar capacity across all investment periods.

Figure 9 illustrates Sweden's initial leadership in wind capacity, with the highest levels during the first investment period. However, Sweden's wind capacity declines significantly after the first period and disappears entirely after the second. Denmark, on the other hand, stands out for its focus on wind energy, with its capacity showing variability and a notable increase in the final period. This aligns with Denmark's reliance on wind as its primary renewable energy source. Norway and Finland show more constant wind capacities over the timeline compared to the significant changes seen in Sweden and Denmark.

In Figure 10, Sweden's dominance in run-of-river hydro capacity is evident, with stable and high levels maintained consistently over the timeline. Norway ranks second with slightly lower but equally stable capacity, while Finland comes third with moderate and constant hydro capacity throughout the timeline. Denmark's hydro capacity, on the other hand, is negligible.

These renewable energy patterns align closely with the earlier BESS capacity analysis. Sweden's significant investment in a diversified renewable portfolio, encompassing solar, wind, and hydro, necessitates the adoption of BESS to enhance grid flexibility and manage variability across different energy sources. In contrast, Norway's reliance on hydro, which inherently provides a natural form of energy storage, minimizes its need for additional BESS investments.

5.3 Energy Mixes across the Nordics

Figures 11 and 12 illustrate the installed capacities by technology type across the Nordic countries, highlighting how each nation's unique resource base and energy strategy shapes its energy mix. Figure 11 focuses on Sweden and Finland, while Figure 12 examines Norway and Denmark.

In Figure 11, Sweden emerges as a leader in solar and hydro run-of-river capacities, with nuclear providing a stable contribution throughout the timeline. The high initial capacity of offshore wind, which disappears after the second period, suggests a strategic shift toward solar and hydro resources. Finland's energy profile complements stable hydro and growing onshore wind with an increasing reliance on gas technologies such as CCGT and OCGT in later periods.



Figure 11: Installed generation capacity (MW) by investment period for Sweden and Finland.

Figure 12 reveals Norway's energy mix, heavily dominated by hydropower, where regulated hydro serves as the largest contributor alongside steady inputs from hydro run-of-river and onshore wind. However, a gradual increase in CCGT capacity in the final period hints at a growing need for additional dispatchable flexibility, which aligns with Norway's minimal yet notable adoption of BESS toward the end of the timeline. Denmark demonstrates a more dynamic capacity mix, characterized by variability in earlier periods and a late increase in onshore wind and OCGT capacities after the third period. The rising role of OCGT reflects the growing flexibility demands of a system heavily reliant on wind energy. The absence of offshore wind in later periods highlights a strategic shift to other technologies to address system requirements.



Figure 12: Installed generation capacity (MW) by investment period for Norway and Denmark.

5.4 Curtailment and Flexibility Needs

Figure 13 illustrates the curtailment levels of VRE for the Nordic countries over the modeled investment periods.

VRE curtailment levels remain lowest in Sweden throughout the periods. This outcome is associated with the substantial installed BESS capacities, as shown in Figure 7, and significant hydro run-of-river capacity within its system. The combination of these resources provides a strong mechanism for absorbing and releasing excess renewable energy, effectively reducing curtailment. These results highlight the high renewable energy utilization achievable through a well-integrated system of storage and flexibility.



Figure 13: Total curtailed production (GWh) by investment period across the Nordic countries.

In contrast, Finland is associated with the highest levels of curtailment among the Nordic countries. Despite the reliance on onshore wind, nuclear, and hydro run-ofriver, the limited BESS capacity appears to hinder the ability to store and manage excess VRE output. This results in significant power curtailment, raising questions as to why the model does not prioritize BESS investments in Finland to mitigate curtailment. The continued use of other flexibility options, such as gas technologies observed in later periods, suggests that the model evaluates these alternatives as a more cost-effective or complementary solution to Finland's energy system. However, the high curtailment levels indicate a potential area for improvement, where targeted BESS investments could enhance renewable integration and reduce energy losses.

Norway, which exhibits moderate curtailment levels, reflects its reliance on hydroelectricity as the primary flexibility resource. The lack of significant BESS investments is attributed to the stability and abundance of hydropower capacity. Hydropower naturally provides flexible management of VRE output. However, the small-scale use of BESS during the final investment period suggests that increasing reliance on renewable energy may create localized challenges that cannot be fully addressed by Norway's hydro resources.

Denmark consistently exhibits minimal curtailment levels throughout the periods despite negligible BESS capacity. This outcome is likely due to the ability to manage energy surpluses and deficits effectively, reducing the reliance on local storage solutions and keeping curtailment levels low.

While these strategies mitigate VRE curtailment to varying degrees, the effectiveness of such measures often hinges on the underlying transmission infrastructure. The next section delves into how transmission capacity and inter-zonal flows influence the region's ability to balance renewable generation and manage variability.

5.5 Transmission Analysis Across the Nordic Region

The transmission results from the model highlight distinct interconnection patterns across the Nordic region, as depicted in Figures 14, 15, 16, and 17.

Figure 14 shows that Finland's primary transmission pathways are the Estonia-Finland and Finland-NO4 connections, which handle significant annual volumes. These interconnections allow the system to manage demand fluctuations without deploying substantial BESS capacity. However, moderate transmission losses, particularly on the Estonia-Finland route, suggest opportunities for improving efficiency.

Sweden's transmission network, illustrated in Figure 15, demonstrates a diverse set of flows, with NO1-SE3 emerging as the highest-volume connection. This strong interconnection supports Sweden's large-scale integration of renewables such as solar, wind onshore, and hydro run-of-river. The model results highlight how Sweden's combination of transmission capacity and significant BESS installations allows for effective internal balancing and low curtailment levels. Additional connections, such as DK1-SE3 and Lithuania-SE4, emphasize the integration of cross-border flows to manage energy variability.



Figure 14: Transmission analysis for Finland showing expected annual transmission volumes and losses.



Figure 15: Transmission analysis for Sweden showing expected annual transmission volumes and losses.

Norway's hydro-dominated energy system is reflected in Figure 16, which highlights high-volume interconnections such as DK1-NO2, NO1-NO3, and Germany-NO2. These interconnections facilitate the export of hydroelectric surpluses, enabling Norway to act as a balancing hub within the region. The model indicates that while hydro resources provide the primary flexibility, a small-scale adoption of BESS in the final investment period could address localized challenges as renewable energy penetration increases.

As shown in Figure 17, Denmark relies on external interconnections, particularly DK1-Germany and DK1-NO2, to manage the variability of its wind-dominated energy system. These connections effectively manage surplus wind generation, limiting the need for local storage. The model also indicates minimal transmission between Denmark's internal zones (DK1 and DK2), suggesting independent operation of these regions with significant dependence on cross-border flows for balancing.



Figure 16: Transmission analysis for Norway showing expected annual transmission volumes and losses.



Figure 17: Transmission analysis for Denmark showing expected annual transmission volumes and losses.

These findings underscore how transmission infrastructure shapes the role and development of BESS across the Nordic region. Strong interconnections often serve as

a substitute for localized storage, as seen in Finland and Denmark, where external flows reduce the immediate need for large-scale BESS deployment. Conversely, countries such as Sweden leverage both transmission and BESS to balance diverse renewable portfolios and manage internal energy flows, emphasizing the complementary nature of these technologies. Norway's reliance on hydro flexibility further illustrates how existing resources can delay or reduce the need for BESS, but it also highlights potential gaps in addressing future challenges as renewable penetration deepens.

5.6 Summary

The results of this study show different approaches to integrating renewable energy and managing flexibility across the Nordic region from 2023 to 2033. The adoption of BESS varies widely, with higher investments in areas that rely on a mix of solar, wind, and hydroelectric power. In these regions, BESS helps reduce power curtailment and improves grid stability, while the existing transmission networks support the balancing of renewable energy across zones.

In areas with large hydroelectric resources, the natural storage provided by hydro systems reduces the need for BESS. Pumped hydro capacities remain consistently high, offering a reliable way to manage energy supply and demand. These regions also benefit from strong transmission links that allow them to export surplus energy efficiently. On the other hand, areas with limited BESS and a focus on specific technologies, such as wind power, experience higher energy curtailment.

Systems that depend heavily on wind energy manage variability through crossborder transmission rather than local storage. These connections are key to balancing energy production and demand. A shift away from offshore wind in some areas, in favor of other technologies, reflects cost-driven optimization within the model, as it prioritizes the most economical solutions based on the given inputs and constraints.

6 Conclusion

This thesis set out to evaluate the role of BESS in supporting the integration of renewable energy across the Nordic energy system, aligning with the ambitious climate and energy targets established by each country. However, the results reveal a more complex and uneven adoption landscape, highlighting the need for region-specific strategies.

The model's findings indicate that BESS adoption is concentrated almost entirely in Sweden, driven by its diverse renewable energy portfolio of solar, wind, and hydro run-of-river, which creates significant grid balancing demands. In contrast, the other Nordic countries show negligible BESS development by 2033. Norway's reliance on its abundant hydro resources, which inherently provide natural flexibility, minimizes the need for BESS investments. Denmark and Finland, despite their renewable energy ambitions, adopt alternative strategies. Notably, Finland's reliance on gas technologies increases in later periods, providing critical flexibility to manage renewable variability. These gas technologies, while effective in the short term, underscore the trade-offs between immediate system stability and long-term decarbonization goals.

The reliance on gas technologies in Finland and the minimal use of BESS suggest an opportunity for a hybrid approach. Targeted BESS investments could address curtailment challenges and improve renewable utilization, complementing existing gas infrastructure. Denmark, on the other hand, leverages strong transmission interconnections to balance variability, maintaining low curtailment levels despite minimal local storage. These diverse approaches reflect the importance of tailoring energy strategies to each country's unique resource base and flexibility needs.

From a policy perspective, these findings provide important insights. Sweden's potential for BESS integration highlights the importance of coupling diversified renewable portfolios with storage investments. Norway's hydro-dominated system showcases the value of leveraging natural flexibility resources, while Finland's and Denmark's strategies emphasize the role of gas and transmission infrastructure in the absence of extensive BESS deployment. Strengthening cross-border transmission networks emerges as a potential priority, enabling the region to collectively manage variability and enhance renewable integration.

Despite these insights, the study is subject to several limitations. The modeling approach relies on static assumptions for renewable generation patterns and energy demand, which may not fully capture future technological advancements or shifts in market dynamics. Additionally, the exclusion of offshore wind farm developments beyond existing clusters and the static treatment of CO_2 pricing reduce the scope for exploring emerging trends in the energy landscape. The model's reliance on temporal aggregation to reduce computational complexity may also obscure finer operational details, particularly during extreme load periods.

Future research should address these limitations by incorporating dynamic modeling of offshore wind developments, exploring variable CO_2 pricing scenarios, and refining the treatment of operational uncertainties. A deeper analysis of the economic and policy incentives required to promote BESS adoption in countries with low uptake, such as Finland and Denmark, could further enhance understanding of the region's

energy transition. Additionally, integrating advanced flexibility options, such as hydrogen storage and demand-side management, could provide a more comprehensive framework for optimizing the Nordic energy system.

While the results provide valuable insights, they diverge from initial expectations that BESS would see widespread adoption across the Nordic region. This outcome underscores the importance of considering country-specific factors in energy planning models. This research contributes to the broader understanding of energy storage in renewable-rich systems, offering a framework for evaluating its role in diverse contexts. By addressing its limitations and building on the recommendations outlined, future work can better support the Nordic region's transition to a sustainable, low-carbon energy system.

References

- [1] European Commission, "Stepping up europe's 2030 climate ambition: Investing in a climate-neutral future for the benefit of our people," Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee, and the Committee of the Regions, Brussels, Tech. Rep. COM(2020) 562 final, September 2020.
- [2] H. Flam and J. Hassler, "Eu climate policy and fit for 55," in *Nordic Economic Policy Review 2023: EU versus National Climate Policies in the Nordics*, F. S. P. B. S. L. G. H. M. L. I. V. M. O. R. G. M. H. Harry Flam, John Hassler *et al.*, Eds. Nordic Council of Ministers, 2023, pp. 5–12.
- [3] A. F. E. A. E. A. Hamdan, C. D. Daudu and S. Sonko, "Next-generation batteries and u.s. energy storage: A comprehensive review: Scrutinizing advancements in battery technology, their role in renewable energy, and grid stability," *World Journal of Advanced Research and Reviews*, vol. 21, no. 01, 2024.
- [4] EnergyTrend, "Battery energy storage market update: 2024 outlook and trends," 2024. [Online]. Available: https://www.energytrend.com/research/20240604-4 7321.html
- [5] J. van Ouwerkerk, H. C. Gils, H. Gardian, M. Kittel, W.-P. Schill, A. Zerrahn, A. Murmann, J. Launer, L. Torralba-Díaz, and C. Bußar, "Impacts of power sector model features on optimal capacity expansion: A comparative study," *Renewable* and Sustainable Energy Reviews, vol. 157, p. 112004, 2022. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1364032121012673
- [6] R. Loulou, G. Goldstein, A. Kanudia, A. Lettila, and U. Remme, "Documentation for the times model, part i," *IEA-ETSAP Documentation*, 2016. [Online]. Available: http://iea-etsap.org/docs/Documentation_for_the_TIMES_Model-P art-I_July-2016.pdf
- [7] L. Schrattenholzer, "The energy supply model message," IIASA, Tech. Rep. RR-81-031, 1981.
- [8] D. Connolly, H. Lund, B. V. Mathiesen, and M. Leahy, "A review of computer tools for analysing the integration of renewable energy into various energy systems," *Applied energy*, vol. 87, no. 4, pp. 1059–1082, 2010.
- [9] S. Backe, C. Skar, P. C. del Granado, O. Turgut, and A. Tomasgard, "Empire: An open-source model based on multi-horizon programming for energy transition analyses," *SoftwareX*, vol. 17, p. 100877, 2022. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2352711021001424
- [10] L. Gacitua, P. Gallegos, R. Henriquez-Auba, Á. Lorca, M. Negrete-Pincetic, D. Olivares, A. Valenzuela, and G. Wenzel, "A comprehensive review on

expansion planning: Models and tools for energy policy analysis," *Renewable and Sustainable Energy Reviews*, vol. 98, pp. 346–360, 2018.

- [11] I. Chernyakhovskiy, M. Joshi, and A. Rose, "Power system planning: advancements in capacity expansion modeling," National Renewable Energy Lab.(NREL), Golden, CO (United States), Tech. Rep., 2021.
- [12] M. Grubb, J. Edmonds, P. Ten Brink, and M. Morrison, "The costs of limiting fossil-fuel co2 emissions: a survey and analysis," *Annual Review of Energy and the environment*, vol. 18, no. 1, pp. 397–478, 1993.
- [13] M. G. Prina, G. Manzolini, D. Moser, B. Nastasi, and W. Sparber, "Classification and challenges of bottom-up energy system models - a review," *Renewable and Sustainable Energy Reviews*, vol. 129, p. 109917, 2020. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S1364032120302082
- [14] N. M. J. P. van Beeck, "Classification of energy models," FEW Research Memorandum, Tech. Rep., 1999.
- [15] C. B. J.-C. Hourcade, M. Jaccard and F. Ghersi, "Hybrid modeling: New answers to old challenges - introduction to the special issue of the energy journal," *The Energy Journal*, vol. 27, no. 2_suppl, pp. 1–11, 2006.
- [16] J. Tomaschek, "Long-term optimization of the transport sector to address greenhouse gas reduction targets under rapid growth. application of an energy system model for gauteng province, south africa," Dec 2013.
- [17] A. Lyden, R. Pepper, and P. G. Tuohy, "A modelling tool selection process for planning of community scale energy systems including storage and demand side management," *Sustainable Cities and Society*, vol. 39, pp. 674–688, 2018. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S2210670717309824
- [18] A. Herbst, F. Toro, F. Reitze, and E. Jochem, "Introduction to energy systems modelling," *Swiss journal of economics and statistics*, vol. 148, pp. 111–135, 2012.
- [19] K. Espegren, S. Damman, P. Pisciella, I. Graabak, and A. Tomasgard, "The role of hydrogen in the transition from a petroleum economy to a low-carbon society," *International Journal of Hydrogen Energy*, vol. 46, no. 45, pp. 23 125–23 138, 2021, hydrogen Separation, Production and Storage. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0360319921015639
- [20] N. Reyseliani and W. W. Purwanto, "Pathway towards 100% renewable energy in indonesia power system by 2050," *Renewable Energy*, vol. 176, pp. 305–321, 2021.

- [21] G. S. Seck, E. Hache, J. Sabathier, F. Guedes, G. A. Reigstad, J. Straus, O. Wolfgang, J. A. Ouassou, M. Askeland, I. Hjorth *et al.*, "Hydrogen and the decarbonization of the energy system in europe in 2050: A detailed model-based analysis," *Renewable and Sustainable Energy Reviews*, vol. 167, p. 112779, 2022.
- [22] N. Nakicenovic, A. Grübler, and A. McDonald, *Global energy perspectives*. Cambridge University Press, 1998.
- [23] N. Nakicenovic, J. Alcamo, G. Davis, B. d. Vries, J. Fenhann, S. Gaffin, K. Gregory, A. Grubler, T. Y. Jung, T. Kram *et al.*, "Special report on emissions scenarios," 2000.
- [24] A. Grübler, N. Nakicenovic, K. Riahi, F. Wagner, G. Fischer, I. Keppo, M. Obersteiner, B. O'Neill, S. Rao, and F. Tubiello, "Integrated assessment of uncertainties in greenhouse gas emissions and their mitigation: Introduction and overview," *Technological Forecasting and Social Change*, vol. 74, no. 7, pp. 873–886, 2007.
- [25] K. Riahi, A. Grübler, and N. Nakicenovic, "Scenarios of long-term socioeconomic and environmental development under climate stabilization," *Technological forecasting and social change*, vol. 74, no. 7, pp. 887–935, 2007.
- [26] E. Shafiei, Y. Saboohi, and M. B. Ghofrani, "Impact of innovation programs on development of energy system: Case of iranian electricity-supply system," *Energy Policy*, vol. 37, no. 6, pp. 2221–2230, 2009.
- [27] International Atomic Energy Agency, "Assessing policy options for increasing the use of renewable energy for sustainable development: Modelling energy scenarios for sichuan, china," International Atomic Energy Agency, Tech. Rep., 2007. [Online]. Available: http: //www.iaea.org/OurWork/ST/NE/Pess/assets/sichuan_case_study.pdf
- [28] A. Galinis, V. Miškinis, J. Vilemas, and D. Tarvydas, "Analyses of energy supply options and security of energy supply in the baltic states," *IAEA Technical Documents*, 2007.
- [29] I. A. E. Agency, Cuba: A Country Profile on Sustainable Energy Development: Non-serial Publications. International Atomic Energy Agency, 2008.
- [30] L. Torralba-Díaz, C. Schimeczek, M. Reeg, G. Savvidis, M. Deissenroth-Uhrig, F. Guthoff, B. Fleischer, and K. Hufendiek, "Identification of the efficiency gap by coupling a fundamental electricity market model and an agent-based simulation model," *Energies*, vol. 13, no. 15, p. 3920, 2020.
- [31] University of Minnesota Libraries Publishing, *Principles of Economics*. University of Minnesota Libraries Publishing, 2023, adapted edition produced through the eLearning Support Initiative under a Creative Commons Attribution-NonCommercial-ShareAlike license. Originally published in 2011.

[Online]. Available: https://open.lib.umn.edu/principleseconomics/chapter/9-1 -perfect-competition-a-model/

- [32] N. Sun, I. Ellersdorfer, and D. J. Swider, "Model-based long-term electricity generation system planning under uncertainty," in 2008 Third International Conference on Electric Utility Deregulation and Restructuring and Power Technologies. IEEE, 2008, pp. 1298–1304.
- [33] C. Schimeczek, K. Nienhaus, U. Frey, E. Sperber, S. Sarfarazi, F. Nitsch, J. Kochems, and A. A. El Ghazi, "Amiris: Agent-based market model for the investigation of renewable and integrated energy systems," *Journal of Open Source Software*, 2023.
- [34] J. Richter, "Dimension-a dispatch and investment model for european electricity markets," EWI working paper, Tech. Rep., 2011.
- [35] M. Bartels, "Cost efficient expansion of district heat networks in germany," Jul 2009.
- [36] A. Prognos, M. Schlesinger, P. D. Dietmar, and C. Lutz, "Energieszenarien für ein energiekonzept der bundesregierung," *Prognos AG, Basel/Köln/Osnabrück*, 2010.
- [37] S. Nagl, M. Fürsch, M. Paulus, J. Richter, J. Trüby, and D. Lindenberger, "Energy policy scenarios to reach challenging climate protection targets in the german electricity sector until 2050," *Utilities Policy*, vol. 19, no. 3, pp. 185–192, 2011, infrastructure Reform in China. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0957178711000324
- [38] S. H. Agricola, Annegret C and I. Zoch, "dena grid study ii. integration of renewable energy sources into the german power supply system until 2020; dena - netzstudie ii. integration erneuerbarer energien in die deutsche stromversorgung bis 2020," Dec 2010.
- [39] M. Paulus and F. Borggrefe, "The potential of demand-side management in energy-intensive industries for electricity markets in germany," *Applied Energy*, vol. 88, no. 2, pp. 432–441, 2011, the 5th Dubrovnik Conference on Sustainable Development of Energy, Water and Environment Systems, held in Dubrovnik September/October 2009. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0306261910000814
- [40] M. Fürsch, C. Golling, M. Nicolosi, R. Wissen, and D. Lindenberger, "European res-e policy analysis-a model based analysis of res-e deployment and its impact on the conventional power market," *Energiewirtschaftliches Institut an der Universität Köln, Köln*, 2010.
- [41] C. Skar, G. Doorman, G. A. Pérez-Valdés, and A. Tomasgard, "A multi-horizon stochastic programming model for the european power system; 2016," 2016.

- [42] AFRY Finland Oy, "Kotimaisten polttoaineiden toimintaympäristö ja käyttöarviot 2028 saakka - Loppuraportti," 2023. [Online]. Available: https://afry.com/sites/default/files/2023-02/kotimaisten_polttoaineiden_toimi ntaymparisto_ja_kayttoarviot_2028_saakka_loppuraportti_8.2.2023.pdf
- [43] A. De Vita, I. Kielichowska, P. Mandatowa, P. Capros, E. Dimopoulou, S. Evangelopoulou, T. Fotiou, M. Kannavou, P. Siskos, G. Zazias *et al.*, "Technology pathways in decarbonisation scenarios," *Tractebel, Ecofys, E3-Modelling: Brussels, Belgium*, 2018.
- [44] Z. E. Platform, "Co2 capture and storage (ccs)," *Recommendations for transitional measures to drive deployment in Europe*, 2013.
- [45] J. Hyvönen, T. Koivunen, and S. Syri, "Possible bottlenecks in clean energy transitions: Overview and modelled effects – case finland," *Journal* of Cleaner Production, vol. 410, p. 137317, 2023. [Online]. Available: https://www.sciencedirect.com/science/article/pii/S0959652623014750
- [46] N. Semkin, M.-M. Ketonen, S. Takamäki, K. Tuominen, H. Mahlamäki, and S. Rättö, "Energiaintensiivisen teollisuuden vihreän siirtymän investointitarpeet ja niiden toteutumisedellytykset," 2023.
- [47] European Network of Transmission System Operators for Electricity, "Statistical factsheet 2023," 2023. [Online]. Available: https://eepublicdownloads.blob.cor e.windows.net/public-cdn-container/clean-documents/Publications/Statistics/F actsheet/entsoe_sfs2023_web.pdf
- [48] G. E. Monitor, "Global wind power tracker, december 2023 release," 2023. [Online]. Available: https://globalenergymonitor.org
- [49] Energiföretagen Sverige, "Energiåret 2023: Tabeller," 2023. [Online]. Available: https://www.energiforetagen.se/496178/globalassets/energiforetagen/statistik /energiaret/2023/energiaret-2023_tabeller.pdf
- [50] G. H. A. R. B. V. H. H. L. R. K. Freeman, C. Frost and I. Pineda, "Our energy, our future: How offshore wind will help europe go carbon-neutral," November 2019. [Online]. Available: https://windeurope.org/intelligence-platform/product/our-energy-our-future/
- [51] T. P. D. L. d. T. H. C. Gils, Y. Scholz and D. Heide, "Integrated modelling of variable renewable energy-based power supply in europe," *Energy*, vol. 123, pp. 173–188, 2017.
- [52] P. V. H. M. F. D. L. D. Tarvydas, S. Politis and J. Badger, Wind potentials for EU and neighbouring countries: Input datasets for the JRC-EU-TIMES model. Publications Office, 2018. [Online]. Available: https://doi.org/10.2760/041705

- [53] B. B. A. C. H. G. M. E. M. L. N. N. B. O.-E. J. D. K. S. D. R. Gómez, J. D. Watterson and K. Treanton, 2006 IPCC Guidelines for National Greenhouse Gas Inventories, Volume 2: Energy. Intergovernmental Panel on Climate Change, 2006, contributing author: R. Quadrelli (IEA). [Online]. Available: https://www.ipcc-nggip.iges.or.jp/public/2006gl/pdf/2_Volume2/V2_2_Ch2 _Stationary_Combustion.pdf
- [54] ENTSO-E TYNDP22, "Demand time series 2025," 2022. [Online]. Available: https://2022.entsos-tyndp-scenarios.eu/download/
- [55] —, "Demand time series 2030," 2022. [Online]. Available: https://2022.entsos-tyndp-scenarios.eu/download/
- [56] Cambridge Economic Policy Associates (CEPA), "Study on the value of lost load in the electricity supply," 2020. [Online]. Available: https://www.acer.europa.eu /sites/default/files/documents/en/Electricity/Infrastructure_and_network%20d evelopment/Infrastructure/Documents/CEPA%20study%20on%20the%20Va lue%20of%20Lost%20Load%20in%20the%20electricity%20supply.pdf
- [57] K. V. L. Pitorac and L. Lia, "Technical review of existing norwegian pumped storage plants," *Energies*, vol. 13, no. 18, p. 4918, 2020. [Online]. Available: https://doi.org/10.3390/en13184918
- [58] M. F. N. Papadakis, C. Fafalakis and D. Katsaprakakis, "A review of pumped hydro storage systems," *Energies*, vol. 16, no. 11, p. 4516, 2023.
- [59] W. Cole and A. Karmakar, Cost Projections for Utility-Scale Battery Storage: 2023 Update. National Renewable Energy Laboratory, 2023. [Online]. Available: https://books.google.fi/books?id=FkZH0AEACAAJ
- [60] W. J. Cole, C. Marcy, V. K. Krishnan, and R. Margolis, "Utility-scale lithium-ion storage cost projections for use in capacity expansion models," in 2016 North American Power Symposium (NAPS), 2016, pp. 1–6.
- [61] Nord Pool Group, "Principles for determining the transfer capacities," March 2023. [Online]. Available: https://www.nordpoolgroup.com/493abd/globalasset s/download-center/tso/principles-for-determining-the-transfer-capacities-202 3-03-23.pdf
- [62] M. Cui, "Introduction to the k-means clustering algorithm based on the elbow method," *Geoscience and Remote Sensing*, vol. 3, pp. 9–16, 2020. [Online]. Available: http://dx.doi.org/10.23977/geors.2020.030102
- [63] NTNU IoT Energy, "Openempire: Open-source european model for power system investment with renewable energy," https://github.com/ntnuiotenergy/O penEMPIRE, 2024.